Author’s response to reviews

Title: Genetic analysis of 62 Chinese families with Duchenne muscular dystrophy and strategies of prenatal diagnosis in a single center

Authors:
Zhengfeng Xu (zhengfeng_xu_nj@163.com)
Jingjing Zhang (18913384682@163.com)
Dingyuan Ma (dingyuanma2002@163.com)
Gang Liu (liugang_516@126.com)
Yuguo Wang (wyg.num1@163.com)
An Liu (fyann1213@163.com)
Li Li (njfybjylily@163.com)
Chunyu Luo (lcy0929@126.com)
Ping Hu (njfybijhuping@163.com)

Version: 2 Date: 16 Oct 2019

Author’s response to reviews:

1. Author list

We note that the order of the authors entered into the submission system differs to that entered into the title page of your manuscript. Please amend the author list in the submission system to match the author list on the title page of your manuscript.

Response: Thanks. We have corrected the author list.

2. Declarations heading

Please ensure that you include a ‘Declarations’ heading and that it is placed after the Abbreviations in your manuscript.
The Declarations section encompasses the sub-sections: Ethics approval and consent to participate, Consent for publication, Availability of data and material, Competing interests, Funding, Authors’ contributions, Acknowledgements.

Response: We have included the "Declarations" section in the manuscript.(Page 13, line 261-291)
3. Role of funding body

Please provide more details regarding the specific role of the funding body in your study. If the funding body only provided the financial means to allow the authors to carry out the study, please state “The funding body played no role in the design of the study and collection, analysis, and interpretation of data and in writing the manuscript” and include this statement in the Funding section of your manuscript. However, if the funding body took part in the design of the study and collection, analysis, and interpretation of data, and the writing of the manuscript, this must be stated in the Competing Interests section of your manuscript.

Response: We have added the funding body. (Page 12, line 272-274)

4. Consent for publication

Please clarify whether consent to publish was written or verbal.

Please also clarify whether written informed consent for publication of the PATIENT’S clinical details and/or clinical images was obtained from the parents of the patients.

Please also clarify if written informed consent for publication of the PARENT’S clinical details and/or clinical images was obtained from the parents of the patients.

Response: We have clarified the consent for publication. (Page 13, line 287-288)

5. Authors’ contributions

Please represent author’s names using their full initials, not their name, and without any titles.

Response: We have represent author's names. (Page 12, line 266-269)

6. Availability of data and materials

Your current statement in the Availability of data and materials section is incorrect. All manuscripts must include an ‘Availability of data and materials’ statement. Data availability statements should include information on where data supporting the results reported in the article can be found including, where applicable, hyperlinks to publicly archived datasets analysed or generated during the study. By data we mean the minimal dataset that would be necessary to interpret, replicate and build upon the findings reported in the article. We recognise it is not always possible to share research data publicly, for instance when individual privacy could be compromised, and in such instances data availability should still be stated in the manuscript along with any conditions for access. Please provide the relevant accession numbers if your data has been deposited into a data bank or the appropriate web links if the data has been uploaded into a repository. If the data has not yet been deposited, please do so and provide the relevant information needed to access it in the ‘Availability of data and materials’ section in your manuscript.

Your data availability statements can take the following form: The datasets generated and/or analysed
during the current study are available in the [NAME] repository, [PERSISTENT WEB LINK TO DATASETS]

Response: We have corrected the "Availability of data and materials". (Page 13, line 277-279) We are uploading the datasets to the ClinVAR database, but it may take a few months. We hope that the datasets in the current study are available in the ClinVAR database in the near future.

7. Figure legends

Please include a “Figure legends” heading above the list of figure legends in your manuscript.

Response: We have added the "Figure legends". (Page 15, line 347)

8. At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Should you wish to respond to these revision requests, please put your responses to the reviewers'/editors’ comments in the Response to Reviewers box in Editorial Manager. Please do not upload a separate letter.

Response: We have uploaded the manuscript, the tables and the figures.